Author’s response to reviews

Title: Implication of Xenobiotic Metabolizing Enzyme gene (CYP2E1, CYP2C19, CYP2D6, mEH and NAT2) Polymorphisms in Breast Carcinoma.

Authors:

Achraf Khedhaier (a_khedhaier@yahoo.fr)
Elham Hassen (elham_tn@yahoo.fr)
Noureddine Bouaouina (noureddine.bouaouina@rns.tn)
Sallouha Gabbouj (sallouha_tn@yahoo.fr)
Slim Ben ahmed (slim.benahmed@rns.tn)
Lotfi Chouchane (lotfi.chouchane@planet.tn)

Version: 4 Date: 13 April 2008

Author’s response to reviews: see over
April 13th, 2008

Dr Maria Kowalczuk
Senior Assistant Editor, BMC-series journals

Manuscript ID : 1231888224155443

Manuscript title: Implication of Xenobiotic Metabolizing Enzyme gene (CYP2E1, CYP2C19, CYP2D6, mEH and NAT2) Polymorphisms in Breast Carcinoma.

Dear Dr Maria Kowalczuk,

We thank you again for all the opportunities that you give us to improve our manuscript. We have made as requested all the “Formatting Changes”. A copy of the revised manuscript and all the figures with all new changes are included.

We removed the "Revised Manuscript # ID : 1231888224155443” from the head of each page. We replaced the title “Materials and Methods” by “Methods”. We presented all text in black. We corrected the references list by adding all the author names. All the tables were put in an A4 portrait page. Finally, the legend and title of the figures were added after the references list.

Please, if you have further requirements, do not hesitate to contact us. Thank you.

Yours sincerely,

Dr Elham HASSEN